Author’s response to reviews

Title: New PCNT Candidate Missense Variant in a Patient with Oral and Maxillofacial Osteodysplasia: A Case Report

Authors:

Ken-Ichi Aoyama (subarumusashi@yahoo.co.jp)
MInoru Kimura (kimura@tokai-u.jp)
Hiroshi Yamazaki (osyamaza@tokai-u.jp)
Masahiro Uchibori (masa09320932@hotmail.co.jp)
Rena Kojima (renakojima86@yahoo.co.jp)
Yuko Osawa (twins.1022.yuko@gmail.com)
Kazuyoshi Hosomichi (khosomic@med.kanazawa-u.ac.jp)
Yoshihide Ota (yotaosrg@yahoo.co.jp)
Masayuki Tanaka (matanaka@tokai-u.jp)
Shiro Yamada (ys060308@tsc.u-tokai.ac.jp)
Gen Nishimura (K-aoyama@taoki-u.jp)

Version: 4 Date: 09 Jul 2019

Author’s response to reviews:

July 09, 2019

Doctor Helena Kuivaniemi

Editor

BMC Medical Genetics
Dear Editor:

I along with my coauthors wish to re-submit a revised manuscript to BMC Medical Genetics, titled “New PCNT Candidate Missense Variant in a Patient with Oral and Maxillofacial Osteodysplasia: A Case Report” (manuscript ID: GTC-D-18-00515R3).

The manuscript has been carefully rechecked and appropriate changes have been made in accordance with the suggestion. The responses to their comments have been prepared and attached below.

1. Consent to participate

We described as ‘We confirm that the patient and patients’ parent have given their written consents for the case report to be published, including their medical data and images’ in Consent for publication section, line 272, page 11.

2. Figure 5a

We have added ‘Figure 5a’ in Candidate gene approach and gene annotation section, line 181, page 8.

3. Formatting

We have deleted CARE checklist.

4. Clean manuscript

The manuscript has been carefully rechecked.

We thank you and the reviewer for your decision and insights, which have enriched the manuscript and produced a more balanced and better account of the research. We hope that the revised manuscript is now suitable for publication in your journal.
I look forward to hearing from you.

Sincerely,

Ken-ichi Aoayama, MD, PhD

Division of Oral and Maxillofacial Surgery, Tokai University School of Medicine

143 Shimokasuya, Isehara, Kanagawa 259-1193, Japan

Telephone number: +81-463-93-1121, ext. 7315233

Fax number: +81-463-96-2892

E-mail: subarumusashi@yahoo.co.jp